Author's response to reviews

Title: Validation of the treatment identification strategy of the HEDIS addiction quality measures: concordance with medical record review

Authors:

Alex HS Harris (Alexander.Harris2@va.gov)
Rachelle N Reeder (Rachelle.Reeder@va.gov)
Laura S Ellerbe (Laura.Ellerbe@va.gov)
Thomas R Bowe (Thomas.Bowe@va.gov)

Version: 4 Date: 23 March 2011

Author's response to reviews:

Major revisions (we require the author to make these changes)

Major revisions (we require the author to make these changes)

Table 5: tables must be created using a Tables tool, and should not be linked files, figures, or generated with tabbed text. Tables must be divided into rows and columns. Please ensure that text alignment in columns is clear and consistent. We recommend the MS Word auto-format template: simple 1. Please do not include vertical lines in your tables.

Response: We addressed this problem as well as the most of the minor concerns. Sorry for these problems.